Author's response to reviews

Title: Association of socioeconomic status with overall overweight and central obesity in men and women: the French Nutrition and Health Survey 2006

Authors:

Michel Vernay (michel.vernay@univ-paris13.fr)
Aurelie Malon (a.malon@smbh.univ-paris13.fr)
Amivi Oleko (amivi.oleko@ined.fr)
Benoit Salanave (benoit.salanave@univ-paris13.fr)
Candice Roudier (c.roudier@invs.sante.fr)
Emmanuelle Szego (e.szego@invs.sante.fr)
Valerie Deschamps (valerie.deschamps@univ-paris13.fr)
Serge Hercberg (s.hercberg@uren.smbh.univ-paris13.fr)
Katia Castetbon (katia.castetbon@univ-paris13.fr)

Version: 5 Date: 22 June 2009

Author's response to reviews: see over
Dear Editor,

Please find enclosed our revised manuscript entitled “Association of socioeconomic status with overall overweight and central obesity in men and women: the French Nutrition and Health Survey 2006”.

We therefore hope that our revised manuscript will comply with your expectancy.

Sincerely Yours,

Dr. Michel Vernay on behalf the authors